**Author's response to reviews**

**Title:** Effect of physical activity intervention based on a pedometer on physical activity level and anthropometric measures after childbirth: a randomized controlled trial

**Authors:**

Masumeh S Maturi (nastaran47@yahoo.com)
Pourandokht Afshary (p.afshary@yahoo.com)
Parvin Abedi (parvinabedi@ymail.com)

**Version:** 7 **Date:** 13 December 2011

**Author's response to reviews:** see over
Dear Editor

Following are what you have requested me to do

1- The abstract is now identical in the submission system and in the manuscript.

2- Reference No 30 has already cited in page 13 line 15, but it was in bracket and I corrected it.

3- I made changes with the tables.

4- I checked the white space around the image, and I think it is minimum, but I may could not understand what exactly do you mean, if you do not mind please check it yourself.